Author's response to reviews

Title: Transcranial direct current stimulation (tDCS) for treatment of major depression during pregnancy: study protocol for a pilot randomized controlled trial

Authors:

Simone Vigod (simone.vigod@wchospital.ca)
Cindy-Lee Dennis (cindylee.dennis@utoronto.ca)
Zafiris Daskalakis (Jeff.Daskalakis@camh.ca)
Kellie Murphy (kmurphy@mtsinaion.ca)
Joel Ray (RayJ@smh.ca)
Tim Oberlander (toberlander@cw.bc.ca)
Sarah Somerton (sarah.somerton@wchospital.ca)
Neesha Hussain-Shamsy (neesha.hussainshamsy@wchospital.ca)
Daniel Blumberger (Daniel.Blumberger@camh.ca)

Version: 4
Date: 3 September 2014

Author's response to reviews: see over
September 3, 2014

RE: Revision Submission for “Transcranial direct current stimulation (tDCS) for treatment of major depression during pregnancy: study protocol for a pilot randomized controlled trial”

Dear Editors,

Thank you for the opportunity to submit minor revisions for our paper entitled: “Transcranial direct current stimulation (tDCS) for treatment of major depression during pregnancy: study protocol for a pilot randomized controlled trial”.

We have addressed the following minor formatting requests in this revision:

1. Please remove the Conclusion section from the Abstract, and incorporate this into the Discussion section.
   We have revised our abstract such that the Conclusion section is incorporated into the Discussion section.

2. Please remove the Conclusions section from the main body of text and incorporate this into the main Discussion section.
   We have removed the conclusions section from the main body of the text and have incorporated this into the main Discussion section.

Thank you so much for the opportunity to submit our revised manuscript to your journal.

Sincerely,

Dr. Simone Vigod, Study Principal Investigator
Scientist, Women’s College Research Institute, Women’s College Hospital
Assistant Professor, Department of Psychiatry (Faculty of Medicine) and Institute of Health Policy, Management and Evaluation (Dalla Lana School of Public Health) University of Toronto; Adjunct Scientist, Institute for Clinical Evaluative Sciences